Author's response to reviews

Title: Comparison of cardiovascular risk factors between Sri Lankans living in Kandy and Oslo

Authors:

Sampath U. B Tennakoon (sampathte@yahoo.com)
Bernadette N Kumar (bernadette.kumar@medisin.uio.no)
Danaseela B Nugegoda (dbnug@yahoo.com)
Haakon E Meyer (h.e.meyer@medisin.uio.no)

Version: 3 Date: 27 October 2010

Author's response to reviews: see over
Cover letter

25.10.2010

Dear Editor,

MS ID: 1449862577382952

Comparison of cardiovascular risk factors between Sri Lankans living in Kandy and Oslo

Thank you very much for accepting our manuscript for publishing. Please find enclosed a revised version of the paper with a point by point description of the changes made.

Sincerely,

Sampath U. B. Tennakoon

(Corresponding author)
Major revisions

Author list: names have been revised as per BMC guidelines.

Minor revisions:

1. The city has been added to the affiliation list
2. All text is in black font now
3. Authors’ contributions have been formatted into a paragraph
4. Reference number 26-the reference has been updated according to details given in obesity reviews journal at: http://onlinelibrary.wiley.com/doi/10.1111/j.1467-789X.2010.00746.x/abstract;jsessionid=2AD7119A6924864617CCA6B9696B5F30.d01t02-on how to site the article concerned. This is an online publication of the article.
5. Table 3 has been re done.